Author's response to reviews

Title: Perspectives on gastroesophageal reflux disease in primary care: the REFLEX study of patient-physician agreement

Authors:

Etienne Dorval (dorval@med.univ-tours.fr)
Jean-Francois Rey (jean-francois.rey@wanadoo.fr)
Christine Soufflet (Christine.Soufflet@astrazeneca.com)
Katarina Halling (khalling@patientreported.com)
Philippe Barthélemy (Philippe.Barthelemy@astrazeneca.com)

Version: 3 Date: 16 December 2010

Author's response to reviews: see over
Dear Dr Chap,

In response to your query about ethical approval, I can confirm that non-interventional observational studies in France do not need to be approved by an ethical committee. Nevertheless, the study was approved by the National Physician Council, the Data Privacy Authority and Scientific Committee for Data Privacy. I hope that this is now clear in the text of the manuscript.

With many thanks for your assistance.

Yours sincerely,

Etienne Dorval
Hôpital Trousseau
Tours 37044
France
Tel: +33 (0)2 47 47 59 16
Fax: +33 (0)2 47 47 84 28
E-mail: dorval@med.univ-tours.fr